Author's response to reviews

Title: Multidrug resistant Kluyvera ascorbata septicemia in an adult patient: a case report

Authors:

Shannon Moonah (smoonah@howard.edu)
Kavita Deonarine (kavitasoleil@yahoo.com)
Clyde Freeman (clcfreeman@Howard.edu)

Version: 4 Date: 26 March 2010

Author's response to reviews:

Dear Independent Editorial Production Team,

Thank you for your suggestions. We have completed all the necessary modifications requested by your team.

1. RE: Major revision
Patient information: please provide the ethnicity of the patient.

We have included the patient’s African-American ethnicity in the manuscript.

2. RE: Minor revisions
Table 1: please edit this so that vertical lines are not visible.

We have made the appropriate adjustments to Table 1.

We have to the best of our ability edited the manuscript and eliminated all possible typographical, spelling or grammatical errors and made our final corrections and revisions.

Sincerely,
Shannon Moonah MD
Howard University Hospital
Washington, DC, USA